Author's response to reviews

Title: Total parathyroidectomy with trace amounts of parathyroid tissue autotransplantation as the treatment of choice for secondary hyperparathyroidism: A Single-Center Experience

Authors:

Qingqing He (heqingqing@yeah.net)
Dayong Zhuang (zdyddy2000@tom.com)
Luming Zheng (oliverzhenglu3@126.com)
Ziyi Fan (miastelo@tom.com)
Peng Zhou (weardzp@126.com)
Jian Zhu (zhujiansdu@126.com)
Songjian Duan (duansongjian123@163.com)
Yanning Li (liyanningn@126.com)
Yanming Ge (jjzyygym@126.com)
Zhen Lv (tengzhoulyzhen@163.com)
Lei Cao (caolei0841020027@163.com)

Version: 2 Date: 4 May 2013

Author's response to reviews: see over
Dear Editor,

Enclosed please find the original article entitled “Total parathyroidectomy with trace amounts of parathyroid tissue autotransplantation as the treatment of choice for secondary hyperparathyroidism: A Single-Center Experience” submitted for your consideration to be published in the “BMC Surgery”. The manuscript has never been prior or duplicate publication or submission elsewhere of any part of the study. This study has not financial or other conflict of interests.

Sincerely yours,

Qingqing He, MD.